Author's response to reviews

Title: Phenotypic and Genetic Characterization of a Cohort of Pediatric Wilson Disease Patients

Authors:

Tawhida Y Abdel Ghaffar (tyghaffar@gmail.com)
Solaf M Elsayed (elsayed683@yahoo.com)
Suzan Elnaghy (suzan_elnaghi@yahoo.com)
Ahmed Shadeed (Ahshadeed@hotmail.com)
Ezzat S Elsobky (elsobkyezzat@hotmail.com)
Hartmut Schmidt (hepar@ukmuenster.de)

Version: 3 Date: 24 May 2011

Author's response to reviews: see over
Dear Editors,

Please find enclosed the revised manuscript titled: “Phenotypic and Genetic Characterization of a Cohort of Pediatric Wilson Disease Patients“. Please find our ‘point-by-point’ description of the changes made:

**Major revisions**

Author list: the list of authors in the manuscript should be written exactly as they are in the submission system, both in style and order.

Done.

Minor

Email address: please include email addresses using the following format:

Done

Highlighting/tracking: Please remove all highlighting and tracking from the manuscript.

Done

Authors’ contributions: please ensure this is formatted into a paragraph rather than a list or in bullet points

Done

Tables- please remove the visible vertical lines from your tables

Done
Figures: It is important for the final layout of the manuscript that the figures are cropped as closely as possible to minimise white space around the image. Our online figure guide contains full details for preparing files for submission and can be viewed here: http://www.biomedcentral.com/info/ifora/figures

Typography: Please take this opportunity to check your manuscript for any typographical errors and to make any final corrections or revisions. This is the final proofing stage for your manuscript, and you will not be able to make any changes after acceptance

Sincerely yours,

Tawhida Y. Abdel Ghaffar, Solaf M. Elsayed, Suzan Elnaghy, Ahmed Shadeed, Ezzat S. Elsobky, Hartmut Schmidt